Author's response to reviews

Title: Compound K, a metabolite of ginseng saponin, induces apoptosis via caspase-8-dependent pathway in HL-60 human leukemia cells

Authors:

Sung-Hee Cho (skstlswltjs@nate.com)
Kyung-Sook Chung (adella76@hanmail.net)
Jung-Hye Choi (apoptosis00@msn.com)
Dong-Hyun Kim (dhkim@khu.ac.kr)
Kyung-Tae Lee (ktlee@khu.ac.kr)

Version: 2 Date: 12 August 2009

Author's response to reviews: see over
July 27, 2009

Editor-in-Chief
BMC Cancer

Dear Dr. Melissa Norton

I declare that data in this paper is original and that it has not been published or accepted for publication, either in whole or in part, in any form. Moreover, I will not be submitted elsewhere.

We would be grateful if the manuscript could be reviewed and considered for publication in BMC Cancer.

Please let me know of your decision at your earliest convenience.

Sincerely yours,

Kyung-Tae Lee, Ph.D.
Department of Pharmaceutical Biochemistry, College of Pharmacy
Kyung-Hee University
Dongdaemun-Ku, Hoegi-Dong 130-701, Seoul
Korea
E-mail: ktlee@khu.ac.kr
Tel: +82-2-9610860
FAX: +82-2-9620860